

Pediatric-inspired protocols in adult acute lymphoblastic leukemia: are the results bearing fruit?

Lori Muffly¹ and Emily Curran²

¹Department of Medicine, Division of Blood and Marrow Transplantation, Stanford University, Stanford, CA; and ²Department of Internal Medicine, Division of Hematology & Oncology, The Vontz Center for Molecular Studies, University of Cincinnati College of Medicine, Cincinnati, OH

Observational findings demonstrating improved survival for younger adults following pediatric, as opposed to adult, acute lymphoblastic leukemia (ALL) regimens have been translated into international, prospective multicenter clinical trials testing the pediatric regimen in young adult ALL. The results of these studies confirm the feasibility of delivering the pediatric regimen in the adult oncology setting and establish the superiority of this approach relative to historical adult cooperative group regimen results. Specific toxicities, including thrombosis, hepatotoxicity, and osteonecrosis, are more prevalent in adults receiving the pediatric regimen relative to young children. Persistent minimal residual disease (MRD) is a strong prognostic indicator in adults receiving the pediatric regimen; sensitive, high-quality MRD evaluation should be performed in all patients receiving these therapies. Incorporation of targeted agents, particularly in the frontline and MRD⁺ setting, will usher in the next era of the pediatric regimen in adult ALL.

Learning Objectives

- Understand the role of the pediatric regimen in adult acute lymphoblastic leukemia through analysis of recently reported results of multicenter prospective clinical trials
- Review the clinical management and toxicities associated with delivering the pediatric regimen in adults

Clinical case, part 1: diagnosis

The patient is a 34-year-old attorney who presented with progressive fatigue, weight loss, and flu-like symptoms. He was found to have cervical lymphadenopathy and a white blood cell (WBC) count of 25 000/µL with 80% circulating blasts; hemoglobin, 8.9 g/dL; and platelets, 28 000/µL. Flow cytometry performed on the peripheral blood demonstrated an abnormal T-cell population expressing CD45 (dim), CD2, cCD3, CD5, CD7, CD8, CD1a, and terminal deoxynucleotidyl transferase, and lacking expression of CD10, CD13, CD19, CD20, CD22, CD33, CD34, and myeloperoxidase. Bone marrow examination confirmed a diagnosis of T-cell acute lymphoblastic leukemia (T-ALL), cortical subtype. Cytogenetic studies demonstrated a normal male karyotype and a targeted next-generation sequencing (NGS) panel detected an activating mutation in NOTCH1. Analysis of the cerebral spinal fluid (CSF) revealed 1 WBC, 5 red blood cells, and no atypical cells. Prior to initiation of therapy, he underwent sperm cryopreservation.

Initial diagnosis and workup

The patient presented in this clinical case received a diagnosis of T-ALL based upon the immunophenotypic analysis of the blasts in

the peripheral blood and bone marrow. In addition to distinguishing B-cell ALL (B-ALL) from T-ALL, a detailed phenotypic analysis is increasingly important at diagnosis and at relapse or progression in order to determine whether targeted antibody-based therapies may be appropriate (eg, rituximab for CD20⁺ B-ALL). In T-ALL, careful determination of the immunophenotype aids in identifying patients with early T-cell precursor ALL (a biologic subtype of T-ALL defined by lack of CD1a and CD8 expression), dim or absent CD5 expression, and expression of 1 or more myeloid- or stem cell– associated antigens.¹ The patient in this clinical case does not have the characteristic phenotype of early T-cell precursor ALL; this is an important clinical determination as this subtype of T-ALL in adults appears to have inferior prognosis and may benefit from intensified consolidation therapy.^{2,3}

Routine cytogenetic analysis remains an essential component of the workup of newly diagnosed ALL in adults. In particular, fluorescent in situ hybridization (FISH) analysis for t(9;22) identifies Philadelphia chromosome–positive (Ph⁺) ALL, and conventional cytogenetics/ FISH may demonstrate additional abnormalities of prognostic importance. In addition, NGS approaches are increasingly being used in the clinic to identify genetic alterations with prognostic and/or predictive significance. For instance, mutations activating NOTCH1 signaling, as in this patient's case, occur in approximately three-quarters of pediatric and young adult T-ALL and are generally associated with a more favorable prognosis.⁴ Although not applicable to this case, a combination of flow cytometry, specialized FISH panels, and RNA-sequencing approaches can be useful in diagnosing the characteristic gene fusions and alterations associated with Ph-like B-ALL, another high-risk biologic subtype.

Conflict-of-interest disclosure: L.M. has been a consultant for Pfizer and Kite and receives research funding from Servier and Adaptive. E.C. declares no competing financial interests.

Off-label drug use: None disclosed.

Table 1. Comparison of "pediatric" and "adult" chemotherapy regimens

Feature	Typical pediatric ALL regimen	Typical adult ALL regimen Lower cumulative dose or no use through therapy		
Asparaginase	Higher cumulative dose			
Vincristine	Higher cumulative dose	Lower cumulative dose		
Intrathecal therapy	Higher cumulative dose	Lower cumulative dose		
Steroid	Higher or similar cumulative dose	Lower or similar cumulative dose		
Anthracycline	Lower cumulative dose	Higher cumulative dose		
Cyclophosphamide	Lower cumulative dose	Higher cumulative dose		
Cranial radiation	Variable/not commonly recommended	Variable		
Hospitalization	Induction only	Often with each cycle		
Adverse effects	More asparaginase-related toxicities	More myelosuppression-related toxicities		

Evaluation of the central nervous system (CNS) with diagnostic lumbar puncture, including cytologic analysis of the CSF, should be routinely performed in patients with ALL. In many treatment protocols, CNS status at time of diagnosis informs specific CNS-directed treatment. The Steinherz/Bleyer algorithm,⁵ as used in the Children's Oncology Group (COG) protocols, can be used to determine CNS involvement should there be peripheral blood contamination of the CSF. Fertility preservation should also be discussed with all younger adults at the time of diagnosis, ideally prior to treatment with cytotoxic chemotherapy.

Clinical case, part 2: frontline therapy

Induction therapy was then initiated following the US Intergroup Cancer and Leukemia Group B (CALGB) 10403 protocol,⁶ a pediatricinspired regimen with multiagent chemotherapy consisting of multiple doses of vincristine, daunorubicin, pegylated asparaginase (peg-asparaginase), and prednisone, in addition to intrathecal cytarabine and methotrexate. The induction was complicated by a significant elevation in liver function studies, including hyperbilirubinemia, which resolved without intervention.

The pediatric regimen as frontline therapy in younger adults Pediatric and medical oncologists have traditionally approached the treatment of ALL using markedly different therapeutic regimens (Table 1). The mainstay of the "pediatric" approach is to use high cumulative doses of nonmyelosuppressive agents such as vincristine, glucocorticoids, and asparaginase, as well as more intensive and prolonged CNS prophylaxis.7-11 Most pediatric treatment protocols are based upon regimens pioneered by the Berlin-Frankfurt-Münster (BFM) study group, which consist of 6 to 9 months of intensive induction/consolidation/delayed intensification therapy, followed by prolonged maintenance. Some pediatric groups in the United States use strategies that differ from BFM-based therapy, but continue to rely on nonmyelosuppressive chemotherapy as a backbone of treatment.9,10 In contrast, "adult" regimens for ALL have historically consisted of intensive use of myelosuppressive agents including daunorubicin, cytarabine, and cyclophosphamide, as well as allogeneic stem cell transplant in first remission (complete remission 1 [CR1]).¹²⁻¹⁴

It has been over a decade since the first observational retrospective studies demonstrated that older adolescents and young adults (AYAs; currently considered to be patients aged 15-39 years old at diagnosis) with ALL fare significantly better following the "pediatric" approach compared with the "adult" approach.^{15,16} This finding, reproduced worldwide, ushered in a novel field of oncology (AYA oncology), and catalyzed a number of prospective clinical trials testing the feasibility and outcome of delivering the pediatric ALL regimen to adults (predominantly younger adults) in the adult oncology setting. In the United States, the adult cancer cooperative groups conducted

CALGB 10403, a prospective phase 2 trial evaluating a pediatric ALL regimen previously used by the COG in high-risk children.^{6,17} From 2007 to 2012, 318 young adults (aged 17-39 years; median, 24 years) with newly diagnosed Ph⁻ B- or T-ALL were enrolled in the clinical trial across 70 adult cancer sites. Compared with the historical 3-year overall survival (OS) of 58% (95% confidence interval [CI], 52%-64%) for young adults treated on previous CALGB ALL clinical trials, the 3-year OS of 73% (95% CI, 68%-78%) for young adults treated on CALGB 10403 was a striking improvement. Similarly, the 3-year disease-free survival (DFS) was significantly higher than the historical rate (66% [95% CI, 60%-72%] vs 48% [95% CI, 41%-55%]). Patients with B- and T-ALL fared equally well and no significant differences in outcomes were observed by decade of age. Interestingly, of the pretreatment variables examined, only obesity (hazard ratio, 1.82; P = .04) and aberrant CRLF2 (hazard ratio, 2.84; P < .001), a commonly involved gene in Ph-like ALL, were independently associated with inferior DFS.

In addition to the US report, several cooperative groups around the world have published results of pediatric regimens administered to adults with ALL in the adult oncology setting (Table 2). The European Group for Research on Adult Acute Lymphoblastic Leukemia (GRAALL) conducted 2 consecutive trials incorporating the pediatric approach to therapy for adults with ALL up to the age of 55 to 60 years.^{18,19} In the larger of the 2 studies, GRAALL-2005, the estimated 2-year event-free survival (EFS) rate of 65.1% (95% CI, 60.5%-69.3%) and the 2-year OS rate of 74.3% (95% CI, 70.0%-78.1%) for patients ≤39 years were similar to the US 10403 results.¹⁸ However, the GRAALL trials also found that toxicity increased and survival dropped in adults over 45 to 55 years, potentially limiting the applicability of this approach to older adults. However, the Dana-Farber Cancer Institute (DFCI) conducted a prospective, multicenter study of a DFCI pediatric consortium regimen with dose-adjusted asparaginase intensification in adults up to age 50 years and found this approach to be tolerable, with similar outcomes in patients across the age spectrum.²⁰ The Nordic and Baltic groups (NOPHO) took a slightly different approach and treated all newly diagnosed ALL patients aged 1 to 45 years on the same pediatric protocol (NOPHO ALL2008), with pediatric and adult oncologists working together to apply this universal protocol.²¹ Although the adolescent and adult groups did not fare quite as well as young children on this protocol, the estimated 5-year EFS (74% \pm 4%) and DFS (78% \pm 3%) for the adult cohort were nonetheless impressive.

It is important to note that none of the prospective cooperative group studies were designed to test the pediatric regimen directly against adult ALL regimens in a randomized fashion, and comparisons currently rely on historical control data. However, it is striking that these multicenter studies have all reported superior outcomes with the pediatric regimen relative to similar, although not identical,

Table 2. Summary of prospective tria	I results of the pediatric regimen	in adolescent and adult ALL
--------------------------------------	------------------------------------	-----------------------------

Trial	Description	No. evaluable	Participant age, y	EFS, % (95% CI)	OS, % (95% Cl)
CALGB 10403 ⁶	Phase 2 single-arm multicenter trial conducted by the US adult cooperative groups using a pediatric regimen based on COG AALL0232	295	17-39	3 y, 59 (54%-65%)	3 y, 73 (68%-78%)
DFCI 01-175 ²⁰	Multicenter trial conducted in the United States and Canada utilizing a pediatric- inspired regimen	92	18-50	4 y, 58 (47%-68%)	4 y, 67 (56%-76%)
GRAALL-2003 ¹⁹	Phase 2 multicenter trial conducted by the GRAALL adult cooperative group across France, Belgium, Switzerland evaluating a pediatric-inspired regimen	225	15-60	3.5 y, 55 (48%-62%)	3.5 y, 60 (53%-66%)
GRAALL-2005 ¹⁸	Multicenter successor trial of GRAALL- 2003; included a randomization to standard or hyperfractionated cyclophosphamide	787	18-59	5 y, 52.2 (48.5%-55.7%)	5 y, 58.5 (54.8%-61.9%)
PETHEMA ALL-96 ⁵⁵	Multicenter trial conducted by the Spanish adult cooperative group utilizing a common pediatric regimen in adolescents and young adults with SR ALL	81	15-30	6 y, 61 (51%-72%)	6 y, 69 (59%-79%)
MDACC ²⁴	Single-center trial conducted by MDACC evaluating an augmented BFM pediatric regimen	106	13-39	Not provided	5 y, 60%
NOPHO ALL2008 ²¹	Multicenter protocol conducted by the Nordic cooperative group evaluating a common pediatric regimen across children, adolescents, and adults	221*	18-45	5 y, 74% \pm 4%†	5 y, 78% ± 3%†

GRALL, Group for Research on Adult Acute Lymphoblastic Leukemia; MDACC, MD Anderson Cancer Center; No., number; NOPHO, Nordic Society of Pediatric Hematology and Oncology; PETHEMA, Programa Español de Tratamiento en Hematología; SR, standard risk.

*A total of 1509 participants were reported, including children and adolescents; the table describes results for the 221 young adults.

†95% CI provided in a different format (as reported in the referenced article).

historical patient populations treated on adult cooperative group protocols. It is hypothesized that the improved results of these modern multicenter adult ALL trials are due to the pediatric regimen itself. It is possible that additional etiologies may also contribute, such as improvements in clinician/investigator compliance with the study regimen, a greater focus on AYAs in oncology and their unique characteristics and needs, and greater patient protocol and medication adherence, although these theories are currently untested. There are also data to suggest that the pediatric regimen may offer additional benefits, such as fewer late occurring health complications, including retained fertility, and less cost to the health care system, although additional investigation and reporting are required on these topics.^{22,23}

In contrast to the study results detailed in the previous sections, in a single-center prospective study, the MD Anderson Cancer Center (MDACC) reported outcomes of a cohort of AYAs treated with an augmented BFM pediatric regimen at their center, and found that outcomes were not significantly different between this approach and their previous hyperfractionated cyclophosphamide, vincristine, Adriamycin, dexamethasone (hyper-CVAD) experience.²⁴ In this report, the 5-year OS for AYAs was 60% with BFM and 60% with hyper-CVAD, and the 5-year continuous complete remission rates were 53% and 55%, respectively. Toxicity profiles differed by approach, with asparaginase-related adverse events (hepatotoxicity, pancreatitis, thrombosis) in the BFM group and significantly greater myelosuppression-associated complications, including grade 3 infections, in the hyper-CVAD group (45% vs 22%; P < .001). Although results with hyper-CVAD in young adults at MDACC are

very encouraging, other groups have not consistently replicated these outcomes and there is concern that this regimen alone, which typically does not incorporate asparaginase, may be insufficient for subtypes of T-ALL.^{25,26} Given that there is unlikely to be a randomized comparison of these approaches in adult ALL, it is advised that adult ALL patients receive frontline therapy at centers experienced with ALL therapies and that the therapeutic regimen reflect the center's expertise. Furthermore, as clinical trial results and real-world outcomes are not always aligned, it is important that population-based studies are conducted in this population to ensure that the clinical trial results are truly translating into improvements in ALL outcomes.

Toxicities associated with the pediatric regimen in adult ALL

Although the pediatric regimen is generally well tolerated in adults with a treatment-related mortality of <5% reported in patients under 45 years of age,^{6,20,21} select regimen-related toxicities are more prevalent in adults relative to young children. The most direct comparison of toxicities across the age spectrum comes from the NOPHO ALL2008 protocol, in which patients between the ages of 1 and 45 years received an identical risk-adapted pediatric ALL regimen.²¹ Although the incidence of asparaginase-related anaphylactic reactions was significantly lower in adults than young children (5.2% vs 14.5%; *P* < .001), thrombosis was much more common in adults than children (17.5% vs 3.6%; *P* < .001), as were pancreatitis (11.3% vs 5.9%; *P* < .001) and osteonecrosis (8.5% vs 2.3%; *P* < .001). Both thrombosis and hepatotoxicity were also noted to be more common toxicities in the CALGB 10403 young adult population than in the COG AALL0232 pediatric population, despite the

use of a similar pediatric regimen.⁶ It was also striking that 30% of the young adults enrolled in CALGB 10403 were obese based upon body mass index, and not only did these patients have inferior DFS, but obesity may have exacerbated the excess hepatic, hyperglycemic, and thrombotic toxicities seen in these patients. As mentioned previously, treatment-related morbidity and mortality also appear to increase when the pediatric regimen is applied to patients over 50 years of age; this approach should therefore be limited to adults younger than 45 to 50 years or in the context of a clinical trial in older adults.^{18,19}

Many of the toxicities observed in the pediatric-based regimens, such as hepatotoxicity, thrombosis, and pancreatitis, have been associated with asparaginase. However, it is well established that asparaginase is an integral component of pediatric-based treatment and several approaches have been incorporated to attempt to mitigate asparaginaserelated toxicities and allow for continued use of asparaginase throughout therapy. Importantly, a peg-asparaginase dose of 2500 IU/m² is used most commonly in pediatric regimens, but pharmacodynamic studies have shown that doses of 1000 to 2000 IU/m² may be sufficient to achieve adequate asparaginase depletion.^{21,27} Therefore, several groups are using lower doses of peg-asparaginase (1000 IU/m²)²¹ and/or capping the dose of peg-asparaginase at 3750 IU (1 vial) and following asparagine depletion or activity levels, which is the strategy in the currently enrolling US AYA ALL adult cooperative group trial A041501 (NCT03150693). Whether these dose adjustments will lead to increased tolerability and decreased toxicity in adult ALL has not yet been established.

Additional strategies to prevent asparaginase-related complications have been described. Although some clinicians may attempt to prevent thrombosis through prophylactic anticoagulation and repletion of hypofibrinogenemia, data are limited to support the efficacy of these approaches.²⁸ Recent reports suggest monitoring and repletion of antithrombin III may decrease the risk of thrombosis in adults treated with asparaginase,^{29,30} but further study is needed. Interest in L-carnitine, an amino acid derivative compound that promotes mitochondrial β-oxidation of long-chain fatty acids, has grown, following case reports of rapid reversal of hepatotoxicity following administration of this compound³¹; a randomized trial of L-carnitine to prevent asparaginaseinduced hepatotoxicity is being planned (W. Stock, University of Chicago, oral communication, 30 June 2019). Finally, although hypersensitivity reactions to asparaginase are not as common in adults as in children, hypersensitivity may be associated with the development of neutralizing antibodies rendering asparaginase ineffective. A premedication strategy of diphenhydramine, hydrocortisone, and acetaminophen prior to each peg-asparaginase dose was introduced into CALGB 10403 after hypersensitivity reactions were seen early on in the study; this intervention reduced the incidence of grade 3/4 hypersensitivity to 4% in subsequent patients.⁶ Therapeutic monitoring of asparaginase activity levels should be strongly considered in patients receiving premedications to ensure adequate asparagine depletion, as the development of silent inactivation due to neutralizing antibodies may not be clinically evident. Patients who experience anaphylaxis following peg-asparaginase should not be rechallenged and should be switched to an Erwinia asparaginase formulation. In contrast, as described in the clinical case in this article, many of the asparaginaseassociated toxicities, including hepatotoxicity, are most commonly selflimiting and should not preclude additional use of this agent.

Clinical case, part 3: response assessment

Following induction and hematopoietic recovery, a bone marrow examination was performed that demonstrated a normocellular marrow with trilineage hematopoiesis and no morphologic evidence of leukemia. Flow cytometry performed at the treating university's hematopoathology laboratory demonstrated no evidence of leukemia (sensitivity of 0.01%) and minimal residual disease (MRD) by NGS demonstrated zero cells remaining of the original leukemia's T-cell receptor (TCR) clonal sequence (sensitivity of 0.0001%). The patient was evaluated by a hematopoietic cell transplantation (HCT) specialist, and a decision was made to pursue consolidation and maintenance therapy following CALGB 10403 rather than HCT in first remission (CR1); he and his 1 full biological sibling did undergo HLA typing and his sibling was haploidentical.

MRD assessment in adults receiving the pediatric regimen

Assessment of MRD response following frontline therapy should be performed in all ALL patients, regardless of patient age or the frontline regimen used. It has been shown repeatedly that the presence of MRD following pediatric induction therapy and persistent MRD following pediatric consolidation is an incredibly important prognostic feature in both children and adults receiving the pediatric regimen.³² The options for MRD evaluation include multiparameter flow cytometry, NGS, and quantitative polymerase chain reaction; expert consensus recommendations conclude that regardless of methodology, a sensitivity of at least 10⁻⁴ is required for adequate MRD assessment in ALL.³³ Relative to multiparameter flow cytometry, detection of ALL MRD by NGS evaluating clonal immunoglobulin heavy chain and/or TCR gene rearrangements provides a higher analytic sensitivity (10^{-6}) and a lower false-negative rate in childhood ALL³⁴; additional investigation is needed to ascertain whether MRD of $<10^{-4}$ offers enhanced prognostic discrimination in adult ALL.

MRD quantification by polymerase chain reaction for clonal immunoglobulin H/TCR rearrangements was performed in 80 young adult patients on the CALGB 10403 study.⁶ Thirty-five (44%) had undetectable MRD at the end of induction; this MRD⁻ group had an excellent 3-year DFS of 85% (95% CI, 74%-98%). Forty-two (53%) had detectable MRD at $>10^{-4}$ and their 3-year DFS of 54% (95% CI, 41%-71%) was significantly worse. Although persistent MRD retained prognostic utility in young adults treated with the pediatric regimen, the rate of end-induction MRD detection was higher in this young adult population than is typically seen in childhood ALL. This suggests that adults with persistent MRD detection may benefit from targeted consolidation therapies to abolish MRD, such as blinatumomab, which was recently approved by the US Food and Drug Administration for MRD⁺ B-ALL in adults.^{35,36} Additional intensification with allogeneic HCT has shown benefit for adults with persistent MRD following pediatric-inspired induction, and should be offered to appropriate adult patients with persistent MRD following pediatric or adult ALL regimens.³⁷ Outcomes following HCT are also superior if patients enter HCT with low or absent MRD38-40; therefore, if feasible, MRD clearance prior to HCT is preferred. Additional investigations are required to determine whether HCT is necessary for all adult patients with CR1, MRD⁺ ALL after clearance of MRD with novel therapies. In the absence of very high-risk features or persistent MRD, allogeneic HCT in CR1 should not routinely be offered to adults receiving the pediatric regimen; therefore, our case patient with T-ALL without high-risk features and in MRD⁻ CR1 was appropriately guided to continue the pediatric regimen through completion of maintenance therapy.

Clinical case, part 4: completion of the pediatric regimen

The patient tolerated the consolidation, interim maintenance, and delayed intensification courses of the pediatric regimen well and was hospitalized only once, briefly, for neutropenic fever; the remainder of his care was delivered in the outpatient setting. He received all scheduled intrathecal therapies and his treatment team recommended against prophylactic cranial irradiation therapy (CRT). He returned to his legal practice at the beginning of maintenance therapy and subsequently completed 3 years of 6-MP, vincristine, methotrexate, prednisone maintenance. Following completion of maintenance, he was referred to a cancer survivorship specialist and received a comprehensive survivorship care plan.

Prophylactic cranial irradiation and the pediatric regimen

Prevention of CNS relapse is an extremely important component of all pediatric ALL regimens, and typically includes aggressive and prolonged delivery of intrathecal as well as systemic therapeutics that cross the blood-brain barrier. Relative to B-ALL, patients with T-ALL have a higher frequency of CNS involvement, are more likely to experience relapse in the CNS, and are more difficult to salvage upon relapse. Therefore, some protocols additionally require prophylactic CRT for patients with T-ALL, but there is no consistent practice across clinical trials. For example, CALGB 10403 included prophylactic CRT for T-ALL to be delivered during the first cycle of maintenance therapy (as well as CRT for patients with CNS involvement at diagnosis); the NOPHO ALL2008 trial omitted CRT altogether.^{6,21} The investigators at St. Jude's Children's Research Hospital have successfully omitted CRT in all children with newly diagnosed ALL, including those with T-ALL and other high-risk features.⁴¹ The CALGB 10403 regimen also included an interim maintenance course with Capizzi-style methotrexate (MTX), which includes escalating IV MTX plus peg-asparaginase. In recent COG studies, Capizzi-style MTX was shown to be potentially superior to high-dose MTX in children and younger adults with T-ALL⁴² but inferior to high-dose MTX in patients with B-ALL.¹⁷ In the absence of randomized data, systematic reviews and meta-analyses evaluating CRT in T-ALL may be informative,43 and decision-making should be personalized and informed by interpretation of available data.

Optimal delivery of pediatric maintenance therapy in adult ALL

The final phase of treatment in the pediatric regimen consists of a prolonged maintenance course, typically 2 to 3 years in duration. During this time, ensuring adherence to therapy and achieving sufficient myelosuppression are critical. It has been shown that <90%adherence to the daily oral mercaptopurine (6MP) that is required throughout maintenance therapy is associated with a significantly increased risk of relapse in children.⁴⁴ Furthermore, an analysis of children and adolescents treated with methotrexate and 6MP maintenance therapy showed that mean WBC count, a surrogate for adherence and lack of adequate myelosuppression, was significantly correlated with risk of relapse.⁴⁵ Adherence to maintenance among young adults treated in adult cancer centers is unknown. As patientreported compliance has been shown to be a poor surrogate for adherence to maintenance therapy, current efforts are under way to assess adherence in AYA ALL using MEMS TrackCap technology (NCT 03150693).

Patients should receive comprehensive survivorship care following completion of therapy. Although analyses of late complications and

survivorship issues are beyond the scope of this article, a thorough review of these topics pertinent to young adult leukemia survivors is available.⁴⁶ Given the lower cumulative cyclophosphamide and doxorubicin dose equivalents with the pediatric regimen as opposed to a traditional adult ALL regimen, it is hypothesized that the incidence of infertility, cardiac complications, and therapy-related myeloid neoplasms may be lower in young adults treated with pediatric-based ALL therapy. Longer follow-up and careful study are required to determine whether the pediatric approach will in fact result in fewer late effects in adult ALL survivors.

The pediatric regimen in adult ALL: future directions

Based upon the data generated from observational studies and the more recent prospective clinical trials outlined in this review, the pediatric regimen has become an international standard of care in newly diagnosed younger adults with Ph⁻ ALL. The management of adult ALL has become increasingly complex, and there appears to be a learning curve associated with delivering the pediatric regimen in adult oncology clinics. Several population-based studies have demonstrated that outcomes for ALL patients are superior following treatment in comprehensive cancer centers.⁴⁷⁻⁴⁹ Although this is not always feasible, efforts should be made to treat ALL patients in centers with disease expertise.

The use of the pediatric regimen in younger adult ALL has been a significant advancement for the field, but the currently reported survival rates are not good enough. The next frontier of ALL therapy includes a personalized approach intended to abolish MRD through the incorporation of novel targeted therapies into the frontline and/ or MRD⁺ settings (see Advani and Copelan⁵⁰ for a comprehensive review on novel agents and MRD-driven approaches in B-ALL). In brief, the US adult Intergroup is currently conducting Alliance 041501 (NCT 03150693), the successor trial to CALGB 10403, which is testing the incorporation of inotuzumab ozogamicin into the frontline setting for young adults with Ph⁻ B-ALL. This trial uses a similar therapeutic backbone to CALGB 10403 and also allows for the incorporation of rituximab in patients with CD20⁺ ALL. Similarly, the ongoing Eastern Cooperative Oncology Group (ECOG) 1910 trial is evaluating blinatumomab in combination with frontline therapy combination chemotherapy in adult ALL (NCT 02003222). The ECOG 1910 trial incorporates components of the AYA regimen but with dose modifications of asparaginase and steroids in patients aged 55 years and older. The pediatric cooperative groups are developing studies of targeted immune and cellular therapies to eradicate MRD, and studies incorporating targeted kinase inhibitors into the pediatric regimen for Ph-like ALL are ongoing (NCT 03571321).

In T-ALL, the COG incorporated nelarabine into frontline pediatric ALL therapy in children and younger adults up to 30 years of age (COG AALL0434) and have shown that this strategy is safe and effective.^{51,52} It is important to note that the dosing and schedule of nelarabine in AALL0434 followed pediatric dosing (650 mg/m² for 5 consecutive days), whereas adult dosing is typically 1.5 mg/m² on days 1, 3, and 5.⁵³ The MDACC also evaluated incorporating nelarabine into hyper-CVAD and found that this was feasible but did not lead to significant improvement in survival relative to historical controls.⁵⁴ Although nelarabine can be incorporated into the pediatric regimen when administered to adults, it has not been proven to improve outcomes in patients over the age of 30 years and the potential toxicities associated with this combination in adults have not been clearly defined. Additional approaches, such as small molecule inhibitors, antibody-based therapies such as daratumumab,

and cellular therapies are being studied for relapsed T-ALL and may ultimately be incorporated into frontline therapy.

Conclusion

The promise of pediatric-inspired regimens in adult ALL has been realized through concerted efforts among multiple international groups to conduct well-designed, prospective trials evaluating the feasibility and outcomes of this approach. The toxicity profile and survival benefits associated with the pediatric regimen demonstrate that this approach can be safely and effectively offered to adults up to the age of 50 years. There is still much room for improvement, but it is clear that incorporation of the pediatric regimen in adult ALL is bearing fruit for numerous adults with ALL around the world.

Correspondence

Lori Muffly, 300 Pasteur Dr, Stanford, CA 94305; e-mail: lmuffly@stanford.edu.

References

- Coustan-Smith E, Mullighan CG, Onciu M, et al. Early T-cell precursor leukaemia: a subtype of very high-risk acute lymphoblastic leukaemia. *Lancet Oncol.* 2009;10(2):147-156.
- Jain N, Lamb AV, O'Brien S, et al. Early T-cell precursor acute lymphoblastic leukemia/lymphoma (ETP-ALL/LBL) in adolescents and adults: a high-risk subtype. *Blood*. 2016;127(15):1863-1869.
- Bond J, Graux C, Lhermitte L, et al. Early response-based therapy stratification improves survival in adult early thymic precursor acute lymphoblastic leukemia: a Group for Research on Adult Acute Lymphoblastic Leukemia study. J Clin Oncol. 2017;35(23):2683-2691.
- Liu Y, Easton J, Shao Y, et al. The genomic landscape of pediatric and young adult T-lineage acute lymphoblastic leukemia. *Nat Genet*. 2017; 49(8):1211-1218.
- Steinherz PG, Seibel NL, Sather H, et al. Treatment of higher risk acute lymphoblastic leukemia in young people (CCG-1961), long-term followup: a report from the Children's Oncology Group [published online ahead of print 28 February 2019]. *Leukemia*. doi:10.1038/s41375-019-0422-z.
- Stock W, Luger SM, Advani AS, et al. A pediatric regimen for older adolescents and young adults with acute lymphoblastic leukemia: results of CALGB 10403. *Blood*. 2019;133(14):1548-1559.
- Gaynon PS, Angiolillo AL, Carroll WL, et al; Children's Oncology Group. Long-term results of the Children's Cancer Group studies for childhood acute lymphoblastic leukemia 1983-2002: a Children's Oncology Group report. *Leukemia*. 2010;24(2):285-297.
- Möricke A, Zimmermann M, Reiter A, et al. Long-term results of five consecutive trials in childhood acute lymphoblastic leukemia performed by the ALL-BFM study group from 1981 to 2000. *Leukemia*. 2010;24(2): 265-284.
- Silverman LB, Stevenson KE, O'Brien JE, et al. Long-term results of Dana-Farber Cancer Institute ALL Consortium protocols for children with newly diagnosed acute lymphoblastic leukemia (1985-2000). *Leukemia*. 2010;24(2):320-334.
- Pui CH, Pei D, Sandlund JT, et al. Long-term results of St Jude Total Therapy Studies 11, 12, 13A, 13B, and 14 for childhood acute lymphoblastic leukemia. *Leukemia*. 2010;24(2):371-382.
- Conter V, Aricò M, Basso G, et al; Associazione Italiana di Ematologia ed Oncologia Pediatrica. Long-term results of the Italian Association of Pediatric Hematology and Oncology (AIEOP) Studies 82, 87, 88, 91 and 95 for childhood acute lymphoblastic leukemia. *Leukemia*. 2010;24(2): 255-264.
- Stock W, Johnson JL, Stone RM, et al. Dose intensification of daunorubicin and cytarabine during treatment of adult acute lymphoblastic leukemia: results of Cancer and Leukemia Group B Study 19802 [published correction appears in *Cancer*. 2014;120(14):2222]. *Cancer*. 2013;119(1):90-98.
- 13. Goldstone AH, Richards SM, Lazarus HM, et al. In adults with standardrisk acute lymphoblastic leukemia, the greatest benefit is achieved from a

matched sibling allogeneic transplantation in first complete remission, and an autologous transplantation is less effective than conventional consolidation/maintenance chemotherapy in all patients: final results of the International ALL Trial (MRC UKALL XII/ECOG E2993). *Blood.* 2008;111(4):1827-1833.

- Rytting ME, Thomas DA, O'Brien SM, et al. Augmented Berlin-Frankfurt-Münster therapy in adolescents and young adults (AYAs) with acute lymphoblastic leukemia (ALL). *Cancer*. 2014;120(23):3660-3668.
- 15. Stock W, La M, Sanford B, et al; Cancer and Leukemia Group B studies. What determines the outcomes for adolescents and young adults with acute lymphoblastic leukemia treated on cooperative group protocols? A comparison of Children's Cancer Group and Cancer and Leukemia Group B studies. *Blood.* 2008;112(5):1646-1654.
- Boissel N, Auclerc MF, Lhéritier V, et al. Should adolescents with acute lymphoblastic leukemia be treated as old children or young adults? Comparison of the French FRALLE-93 and LALA-94 trials. J Clin Oncol. 2003;21(5):774-780.
- Larsen EC, Devidas M, Chen S, et al. Dexamethasone and high-dose methotrexate improve outcome for children and young adults with highrisk B-acute lymphoblastic leukemia: a report from Children's Oncology Group study AALL0232. J Clin Oncol. 2016;34(20):2380-2388.
- Huguet F, Chevret S, Leguay T, et al; Group of Research on Adult ALL (GRAALL). Intensified therapy of acute lymphoblastic leukemia in adults: report of the randomized GRAALL-2005 clinical trial. *J Clin Oncol.* 2018;36(24):2514-2523.
- Huguet F, Leguay T, Raffoux E, et al. Pediatric-inspired therapy in adults with Philadelphia chromosome-negative acute lymphoblastic leukemia: the GRAALL-2003 study [published correction appears in *J Clin Oncol.* 2009;27(15):2574]. *J Clin Oncol.* 2009;27(6):911-918.
- DeAngelo DJ, Stevenson KE, Dahlberg SE, et al. Long-term outcome of a pediatric-inspired regimen used for adults aged 18-50 years with newly diagnosed acute lymphoblastic leukemia. *Leukemia*. 2015;29(3):526-534.
- Toft N, Birgens H, Abrahamsson J, et al. Results of NOPHO ALL2008 treatment for patients aged 1-45 years with acute lymphoblastic leukemia. *Leukemia*. 2018;32(3):606-615.
- Siegel SE, Stock W, Johnson RH, et al. Pediatric-inspired treatment regimens for adolescents and young adults with Philadelphia chromosomenegative acute lymphoblastic leukemia: a review. *JAMA Oncol.* 2018;4(5): 725-734.
- Siegel SE, Advani A, Seibel N, et al. Treatment of young adults with Philadelphia-negative acute lymphoblastic leukemia and lymphoblastic lymphoma: hyper-CVAD vs. pediatric-inspired regimens. *Am J Hematol.* 2018;93(10):1254-1266.
- 24. Rytting ME, Jabbour EJ, Jorgensen JL, et al. Final results of a single institution experience with a pediatric-based regimen, the augmented Berlin-Frankfurt-Münster, in adolescents and young adults with acute lymphoblastic leukemia, and comparison to the hyper-CVAD regimen. *Am J Hematol.* 2016;91(8):819-823.
- Alabdulwahab AS, Elsayed HG, Sherisher MA, Zeeneldin A, Alghamdi K, Elbjeirami WM. The Dana Farber consortium protocol (DFCP) vs. classic hyper-CVAD for treatment of acute lymphoblastic leukemia in patients <50 Y. Single institution experience. *Leuk Res.* 2017;60:58-62.
- Kozlowski P, Åström M, Ahlberg L, et al; Swedish Adult ALL Group. High relapse rate of T cell acute lymphoblastic leukemia in adults treated with hyper-CVAD chemotherapy in Sweden. *Eur J Haematol.* 2014; 92(5):377-381.
- Douer D, Aldoss I, Lunning MA, et al. Pharmacokinetics-based integration of multiple doses of intravenous pegaspargase in a pediatric regimen for adults with newly diagnosed acute lymphoblastic leukemia. *J Clin Oncol.* 2014;32(9):905-911.
- Stock W, Douer D, DeAngelo DJ, et al. Prevention and management of asparaginase/pegasparaginase-associated toxicities in adults and older adolescents: recommendations of an expert panel. *Leuk Lymphoma*. 2011;52(12):2237-2253.
- Grose K, Sen JM, Morris AL, et al. Decrease of asparaginase induced thrombotic events with antithrombin III monitoring and repletion [abstract]. *Blood.* 2018;132(suppl 1):1381.

- Farrell K, Fyfe A, Allan J, Tait RC, Leach M. An antithrombin replacement strategy during asparaginase therapy for acute lymphoblastic leukemia is associated with a reduction in thrombotic events. *Leuk Lymphoma*. 2016;57(11):2568-2574.
- Rausch CR, Paul S, Marx KR, et al. L-carnitine and vitamin B complex for the treatment of pegasparaginase-induced hyperbilirubinemia. *Clin Lymphoma Myeloma Leuk.* 2018;18(5):e191-e195.
- 32. Berry DA, Zhou S, Higley H, et al. Association of minimal residual disease with clinical outcome in pediatric and adult acute lymphoblastic leukemia: a meta-analysis. *JAMA Oncol.* 2017;3(7):e170580.
- 33. Short NJ, Jabbour E, Albitar M, et al. Recommendations for the assessment and management of measurable residual disease in adults with acute lymphoblastic leukemia: a consensus of North American experts. *Am J Hematol.* 2019;94(2):257-265.
- Wood B, Wu D, Crossley B, et al. Measurable residual disease detection by high-throughput sequencing improves risk stratification for pediatric B-ALL. *Blood*. 2018;131(12):1350-1359.
- 35. Curran E, Stock W. Taking a "BiTE out of ALL": blinatumomab approval for MRD-positive ALL. *Blood*. 2019;133(16):1715-1719.
- Gökbuget N, Dombret H, Bonifacio M, et al. Blinatumomab for minimal residual disease in adults with B-cell precursor acute lymphoblastic leukemia. *Blood.* 2018;131(14):1522-1531.
- Dhédin N, Huynh A, Maury S, et al; GRAALL group. Role of allogeneic stem cell transplantation in adult patients with Ph-negative acute lymphoblastic leukemia. *Blood*. 2015;125(16):2486-2496.
- Sutton R, Shaw PJ, Venn NC, et al. Persistent MRD before and after allogeneic BMT predicts relapse in children with acute lymphoblastic leukaemia. *Br J Haematol.* 2015;168(3):395-404.
- Leung W, Pui CH, Coustan-Smith E, et al. Detectable minimal residual disease before hematopoietic cell transplantation is prognostic but does not preclude cure for children with very-high-risk leukemia. *Blood*. 2012; 120(2):468-472.
- Logan AC, Vashi N, Faham M, et al. Immunoglobulin and T cell receptor gene high-throughput sequencing quantifies minimal residual disease in acute lymphoblastic leukemia and predicts post-transplantation relapse and survival. *Biol Blood Marrow Transplant*. 2014;20(9):1307-1313.
- Pui CH, Campana D, Pei D, et al. Treating childhood acute lymphoblastic leukemia without cranial irradiation. *N Engl J Med.* 2009;360(26): 2730-2741.
- 42. Winter SS, Dunsmore KP, Devidas M, et al. Improved survival for children and young adults with T-lineage acute lymphoblastic leukemia: results from the Children's Oncology Group AALL0434 methotrexate randomization. J Clin Oncol. 2018;36(29):2926-2934.
- 43. Kelly MJ, Pauker SG, Parsons SK. Using nonrandomized studies to inform complex clinical decisions: the thorny issue of cranial radiation

therapy for T-cell acute lymphoblastic leukemia. *Pediatr Blood Cancer*. 2015;62(5):790-797.

- 44. Bhatia S, Landier W, Hageman L, et al. 6MP adherence in a multiracial cohort of children with acute lymphoblastic leukemia: a Children's Oncology Group study. *Blood*. 2014;124(15):2345-2353.
- 45. Schmiegelow K, Heyman M, Gustafsson G, et al; Nordic Society of Paediatric Haematology and Oncology (NOPHO). The degree of myelosuppression during maintenance therapy of adolescents with B-lineage intermediate risk acute lymphoblastic leukemia predicts risk of relapse. *Leukemia*. 2010;24(4):715-720.
- Baker KS, Syrjala KL. Long-term complications in adolescent and young adult leukemia survivors. *Hematology Am Soc Hematol Educ Program*. 2018;2018:146-153.
- Muffly L, Alvarez E, Lichtensztajn D, Abrahão R, Gomez SL, Keegan T. Patterns of care and outcomes in adolescent and young adult acute lymphoblastic leukemia: a population-based study. *Blood Adv.* 2018; 2(8):895-903.
- Muffly L, Li Q, Alvarez E, et al. Hematopoietic cell transplantation in young adult acute lymphoblastic leukemia: a United States populationlevel analysis. J Adolesc Young Adult Oncol. 2019;8(3):254-261.
- 49. Wolfson J, Sun CL, Wyatt L, Stock W, Bhatia S. Adolescents and young adults with acute lymphoblastic leukemia and acute myeloid leukemia: impact of care at specialized cancer centers on survival outcome. *Cancer Epidemiol Biomarkers Prev.* 2017;26(3):312-320.
- Advani AS, Copelan EA. Navigating the nexus of MRD and novel agents in ALL. *Hematology Am Soc Hematol Educ Program*. 2019;2019:9-16.
- 51. Winter SS, Dunsmore KP, Devidas M, et al. Safe integration of nelarabine into intensive chemotherapy in newly diagnosed T-cell acute lymphoblastic leukemia: Children's Oncology Group study AALL0434. *Pediatr Blood Cancer*. 2015;62(7):1176-1183.
- Dunsmore KP, Winter S, Devidas M, et al. COG AALL0434: a randomized trial testing nelarabine in newly diagnosed T-cell malignancy [abstract]. J Clin Oncol. 2018;36(suppl 15):10500.
- 53. DeAngelo DJ, Yu D, Johnson JL, et al. Nelarabine induces complete remissions in adults with relapsed or refractory T-lineage acute lymphoblastic leukemia or lymphoblastic lymphoma: Cancer and Leukemia Group B study 19801. *Blood.* 2007;109(12):5136-5142.
- 54. Abaza Y, M Kantarjian H, Faderl S, et al. Hyper-CVAD plus nelarabine in newly diagnosed adult T-cell acute lymphoblastic leukemia and T-lymphoblastic lymphoma. *Am J Hematol.* 2018;93(1):91-99.
- 55. Ribera JM, Oriol A, Sanz MA, et al. Comparison of the results of the treatment of adolescents and young adults with standard-risk acute lymphoblastic leukemia with the Programa Español de Tratamiento en Hematología pediatric-based protocol ALL-96. *J Clin Oncol.* 2008; 26(11):1843-1849.